Author's response to reviews

Title: Development and evaluation of a hand held computer based on-call pack for health protection out of hours duty: A pilot study

Authors:

Ibrahim Abubakar (i.abubakar@uea.ac.uk)
Christopher J Williams (kitwilliams@nhs.net)
Marian McEvoy (m.mcevoy@herts-ha.nhs.uk)

Version: 3 Date: 5 April 2005

Author's response to reviews:

Formatting changes requested
---------------------------------

Dear Editor,
We have made the following formatting changes as requested.

Author list - The list of authors has been written in an identical fashion in the manuscript and online submission system.
Country - Name of country now included in affiliation details.
Keywords section - Section removed.
Introduction - Renamed 'Background'.
Method - Renamed 'Methods'.
Result - Renamed "Results".

Many thanks
Yours sincerely,

Dr I Abubakar